% The following UNIX script will create a hard copy version of the
% REDUCE bibliography from this file and the bibliography file bibl.bib.
%
%       #  Make REDUCE bibliography
%       latex bibl
%       bibtex bibl
%       latex bibl
%       latex bibl
%
\documentstyle [11pt]{article}
\def\thebibliography#1{\section*{}\list
 {[\arabic{enumi}]}{\settowidth\labelwidth{[#1]}\leftmargin\labelwidth
 \advance\leftmargin\labelsep
 \usecounter{enumi}}
 \def\newblock{\hskip .11em plus .33em minus .07em}
 \sloppy\clubpenalty4000\widowpenalty4000
 \sfcode`\.=1000\relax}
\textwidth 6.6in\textheight 9in\columnwidth\textwidth
\hoffset-2cm
\begin{document}
\setcounter{page}{0}
\title{REDUCE Bibliography}
\author{Anthony C. Hearn\\
RAND \\
Santa Monica CA 90407-2138  \vspace {.5cm} \\
May 1991}
\date{}
\maketitle
\vspace{1cm}
This document contains a list of all known references to REDUCE.  It no doubt
contains errors and omissions.  Please report these by regular mail
(preferably in BibTeX format) to the REDUCE Secretary, RAND, P.O. Box 2138,
Santa Monica CA 90407-2318, or by electronic mail to reduce@rand.org.
\begin{center}
\vspace{9.0cm}
RAND Publication CP162 (Rev. 5/91) \vspace*{.5cm} \\
Copyright \copyright 1991 RAND.  All rights reserved.
\end{center}
\newpage
\voffset-2.5cm

\nocite{Abbott:85}
\nocite{Abbott:86}
\nocite{Abbott:87}
\nocite{Abbott:87a}
\nocite{Abbott:88}
\nocite{Abbott:88a}
\nocite{Abbott:89}
\nocite{Abbott:89a}
\nocite{Abdali:88}
\nocite{Abiezzi:83}
\nocite{Adamchik90}
\nocite{Adams:83}
\nocite{Adkins:83}
\nocite{Adkins:83a}
\nocite{Adkins:85}
\nocite{Aguilera-Navarro:87}
\nocite{Akselrod:90}
\nocite{Aldins:69}
\nocite{Alekseev:86}
\nocite{Alekseev:87}
\nocite{Alekseev:87a}
\nocite{Alfeld:82}
\nocite{Amirkhanov:87}
\nocite{Antweiler:89}
\nocite{Appelquist:70}
\nocite{Arbuzov:86}
\nocite{Aso:81}
\nocite{Atherton:73}
\nocite{Aurenche:84}
\nocite{Aurenche:84a}
\nocite{Autin:89}
\nocite{Baekler:84}
\nocite{Baekler:84a}
\nocite{Baekler:86}
\nocite{Baekler:87}
\nocite{Baekler:87a}
\nocite{Baekler:87b}
\nocite{Baekler:88}
\nocite{Baekler:88a}
\nocite{Baekler:88b}
\nocite{Bahrdt:90}
\nocite{Bajla:78}
\nocite{Balian:78}
\nocite{Baker:81}
\nocite{Bark:78}
\nocite{Barthes-Biesel:73}
\nocite{Barton:72}
\nocite{Bateman:86}
\nocite{Bennett}
\nocite{Berends:81}
\nocite{Berkovich:89}
\nocite{Berkovich:90}
\nocite{Berman:63}
\nocite{Bessis:85}
\nocite{Billoire:78}
\nocite{Biro:86}
\nocite{Biro:87}
\nocite{Birrell:77}
\nocite{Biswas:75}
\nocite{Bittencourt:90}
\nocite{Boege:86}
\nocite{Bogdanova:88}
\nocite{Bordoni:81}
\nocite{Bowyer:87}
\nocite{Boyd:78}
\nocite{Brackx:87}
\nocite{Brackx:87a}
\nocite{Brackx:89}
\nocite{Bradford:86}
\nocite{Bradford:88}
\nocite{Bradford90}
\nocite{Broadhurst:85}
\nocite{Brodsky:62}
\nocite{Brodsky:67}
\nocite{Brodsky:69}
\nocite{Brodsky:70}
\nocite{Brodsky:71}
\nocite{Brodsky:72}
\nocite{Brodsky:72a}
\nocite{Brodsky:72b}
\nocite{Brodsky:73}
\nocite{Broughan:82}
\nocite{Brown:79}
\nocite{Bryan-Jones:87}
\nocite{Calmet:72}
\nocite{Calmet:72a}
\nocite{Calmet:74}
\nocite{Calmet:83}
\nocite{Campbell:67}
\nocite{Campbell:68}
\nocite{Campbell:70}
\nocite{Campbell:70a}
\nocite{Campbell:74}
\nocite{Campbell:87}
\nocite{Caprasse:84}
\nocite{Caprasse:85}
\nocite{Caprasse:86}
\nocite{Caprasse:86a}
\nocite{Caprasse:88}
\nocite{Caprasse:89a}
\nocite{Caprasse:90}
\nocite{Caprasse:90a}
\nocite{Carlson:80}
\nocite{Carroll:73}
\nocite{Carroll:75}
\nocite{Cejchan}
\nocite{Chaffy:88}
\nocite{Chinnick:86}
\nocite{Cline:90}
\nocite{Cohen:76}
\nocite{Cohen:76a}
\nocite{Cohen:77}
\nocite{Cohen:79}
\nocite{Cohen:84}
\nocite{Cohen:89}
\nocite{Connor:84}
\nocite{Connor:84a}
\nocite{Conwell:84}
\nocite{Cowan:79}
\nocite{Cung:75}
\nocite{Darbaidze:86}
\nocite{Darbaidze:86a}
\nocite{Darbaidze:88}
\nocite{Darbaidze:89}
\nocite{Dautcourt:79}
\nocite{Dautcourt:80}
\nocite{Dautcourt:81}
\nocite{Dautcourt:83}
\nocite{Davenport:81}
\nocite{Davenport:82}
\nocite{Davenport:82a}
\nocite{Davenport:85}
\nocite{Davenport:88}
\nocite{Davenport:88a}
\nocite{Della-Dora:81}
\nocite{Della-Dora:84}
\nocite{Della-Dora:85}
\nocite{Demaret:89}
\nocite{DeMenna:87}
\nocite{Demichev:85}
\nocite{Demichev:86}
\nocite{deRop:88}
\nocite{DeVos:89}
\nocite{Dewar:89}
\nocite{Dhar:85}
\nocite{Dicrescenzo:85}
\nocite{Diver}
\nocite{Diver:86}
\nocite{Diver:88}
\nocite{Diver:88a}
\nocite{Diver:91}
\nocite{Dorfi:85}
\nocite{Dorizzi:86}
\nocite{dosSantos:85}
\nocite{dosSantos:87}
\nocite{dosSantos:87a}
\nocite{dosSantos:88a}
\nocite{dosSantos:90}
\nocite{Drska:90}
\nocite{Dubowsky:75}
\nocite{Dudley:89}
\nocite{Dufner:69}
\nocite{Dulyan:87}
\nocite{Duncan:86}
\nocite{Duval:87}
\nocite{Earles:70}
\nocite{Eastwood:87}
\nocite{Eastwood:87a}
\nocite{Eastwood:91}
\nocite{Edelen:81}
\nocite{Edelen:82}
\nocite{Edneral:89}
\nocite{Eisenberger:90}
\nocite{Eissfeller:86}
\nocite{Eitelbach:73}
\nocite{Eleuterio:82}
\nocite{Eliseev:85}
\nocite{Elishakoff:87}
\nocite{Elishakoff:87a}
\nocite{Esteban:90}
\nocite{Falck:89}
\nocite{Fazio:84}
\nocite{Fedorova:87}
\nocite{Fedorova:87a}
\nocite{Feldmar:86}
\nocite{Feuillebois:84}
\nocite{Fitch:73}
\nocite{Fitch:81}
\nocite{Fitch:83}
\nocite{Fitch:85}
\nocite{Fitch:85a}
\nocite{Fitch:87}
\nocite{Fitch:87a}
\nocite{Fitch:89}
\nocite{Fitch:89a}
\nocite{Fitch90}
\nocite{Fitch:90a}
\nocite{Flatau:86}
\nocite{Flath:86}
\nocite{Fleischer:71}
\nocite{Fleischer:73}
\nocite{Fleischer:75}
\nocite{Fogelholm:82}
\nocite{Foster:89}
\nocite{Fox:71}
\nocite{Fox:74}
\nocite{Franceschetti:85}
\nocite{Freire:88}
\nocite{Freire:89}
\nocite{Frick:82}
\nocite{Fujimoto:84}
\nocite{Fuzio:85}
\nocite{Gaemers}
\nocite{Gaemers:78}
\nocite{Ganzha:89}
\nocite{Ganzha90}
\nocite{Ganzha90a}
\nocite{Garavaglia}
\nocite{Garavaglia:80}
\nocite{Garavaglia:84}
\nocite{Garcia:86}
\nocite{Garrad:86}
\nocite{Gastmans:79}
\nocite{Gatermann:90}
\nocite{Gatermann90a}
\nocite{Gatermann:90b}
\nocite{Gates:85}
\nocite{Gates:85a}
\nocite{Gates:85b}
\nocite{Gates:85c}
\nocite{Gates:86}
\nocite{Gebauer:85}
\nocite{Gebauer:88}
\nocite{George:68}
\nocite{Gerdt:80}
\nocite{Gerdt:80a}
\nocite{Gerdt:80b}
\nocite{Gerdt:85}
\nocite{Gerdt:85a}
\nocite{Gerdt:85b}
\nocite{Gerdt:85c}
\nocite{Gerdt:86}
\nocite{Gerdt:87}
\nocite{Gerdt:87a}
\nocite{Gerdt:89}
\nocite{Gerdt:89a}
\nocite{Gerdt:89b}
\nocite{Gerdt90}
\nocite{Gerdt90a}
\nocite{Gerdt:90b}
\nocite{Gerdt:90c}
\nocite{Gervois:74}
\nocite{Gladd:82}
\nocite{Gladkih:83}
\nocite{Gladkih:84}
\nocite{Goldman:89}
\nocite{Golley}
\nocite{Good:75}
\nocite{Goto:77}
\nocite{Goto:78}
\nocite{Gould:84}
\nocite{Gragert:81}
\nocite{Grammaticos}
\nocite{Grammaticos:78}
\nocite{Grammaticos:85}
\nocite{Greenland:84}
\nocite{Grimm}
\nocite{Griss:74}
\nocite{Griss:74a}
\nocite{Griss:75}
\nocite{Griss:76}
\nocite{Griss:76a}
\nocite{Griss:77}
\nocite{Griss:77a}
\nocite{Griss:78}
\nocite{Griss:78a}
\nocite{Griss:79}
\nocite{Griss:79a}
\nocite{Grozin:88}
\nocite{Grozin:88a}
\nocite{Grozin:88b}
\nocite{Gunion:72}
\nocite{Gunion:73}
\nocite{Gunion:85}
\nocite{Hadinger:87}
\nocite{Handy:87}
\nocite{Harper:87}
\nocite{Harper:89}
\nocite{Harper:89a}
\nocite{Harrington:77}
\nocite{Harrington:77a}
\nocite{Harrington:79}
\nocite{Harrington:79a}
\nocite{Hasenfratz:80}
\nocite{Hearn:68}
\nocite{Hearn:69}
\nocite{Hearn:71}
\nocite{Hearn:71a}
\nocite{Hearn:72}
\nocite{Hearn:72a}
\nocite{Hearn:72b}
\nocite{Hearn:73}
\nocite{Hearn:73a}
\nocite{Hearn:74}
\nocite{Hearn:74a}
\nocite{Hearn:76}
\nocite{Hearn:76a}
\nocite{Hearn:76b}
\nocite{Hearn:77}
\nocite{Hearn:78}
\nocite{Hearn:79}
\nocite{Hearn:79a}
\nocite{Hearn:80}
\nocite{Hearn:81}
\nocite{Hearn:81a}
\nocite{Hearn:82}
\nocite{Hearn:82a}
\nocite{Hearn:85}
\nocite{Hearn:86}
\nocite{Hearn:87}
\nocite{Hermann:83}
\nocite{Hess:84}
\nocite{Hettich:77}
\nocite{Hietarinta:83}
\nocite{Hietarinta:83a}
\nocite{Hietarinta:84}
\nocite{Hietarinta:84a}
\nocite{Hietarinta:84b}
\nocite{Hietarinta:85}
\nocite{Hietarinta:87}
\nocite{Hietarinta:87a}
\nocite{Hietarinta:87b}
\nocite{Hietarinta:87c}
\nocite{Hietarinta:88}
\nocite{Hietarinta:89}
\nocite{Hirota:89}
\nocite{Horowitz:75}
\nocite{Horwitz:83}
\nocite{Hughes:90}
\nocite{Hulshof:84}
\nocite{Hulshof:85}
\nocite{Hulshof:81}
\nocite{Hulshof:83}
\nocite{Husberg:81}
\nocite{Idesawa:77}
\nocite{Ilyin:87}
\nocite{Ilyin:89}
\nocite{Inada:80}
\nocite{Ioakimidis:90}
\nocite{Ioakimidis:90a}
\nocite{Ito:85}
\nocite{Ito:85a}
\nocite{Ito:88}
\nocite{Ito:90}
\nocite{Ito:90a}
\nocite{Jansen:86}
\nocite{Janssen:87}
\nocite{Jeffrey:84}
\nocite{Kadlecsik:88}
\nocite{Kagan:85}
\nocite{Kagan:88}
\nocite{Kahn:69}
\nocite{Kamal:81}
\nocite{Kamel:69}
\nocite{Kamel:69a}
\nocite{Kamel:78}
\nocite{Kanada:81}
\nocite{Kanada:75}
\nocite{Kaneko:89}
\nocite{Kaps:85}
\nocite{Karr:85}
\nocite{Katsura:85}
\nocite{Kauffman:73}
\nocite{Kazasov:87}
\nocite{Keady:85}
\nocite{Keener:83}
\nocite{Keener:85}
\nocite{Keener:89}
\nocite{Keener:90}
\nocite{Kendall:88}
\nocite{Kendall:89}
\nocite{Kendall:89a}
\nocite{Kendall:90}
\nocite{Kerner:75}
\nocite{Kersten:83}
\nocite{Kersten:84}
\nocite{Kersten:86}
\nocite{Kersten:86a}
\nocite{Kersten:86b}
\nocite{Killalea:80}
\nocite{Kinoshita:72}
\nocite{Kinoshita:73}
\nocite{Kitatani:86}
\nocite{Kobayashi:84}
\nocite{Kobayashi:88}
\nocite{Kodaira:85}
\nocite{Koh:82}
\nocite{Koelbig:81}
\nocite{Koelbig:81b}
\nocite{Koelbig:82}
\nocite{Koelbig:82a}
\nocite{Koelbig:83}
\nocite{Koelbig:83a}
\nocite{Koelbig:84}
\nocite{Koelbig:84a}
\nocite{Koelbig:84b}
\nocite{Koelbig:85}
\nocite{Koelbig:85a}
\nocite{Koelbig:86}
\nocite{Kolar:90}
\nocite{Kornyak:87}
\nocite{Kotorynski:86}
\nocite{Krack:82}
\nocite{Kraus:73}
\nocite{Kredel:88}
\nocite{Kruse:83}
\nocite{Kryukov}
\nocite{Kryukov:84}
\nocite{Kryukov:85}
\nocite{Kryukov:85a}
\nocite{Kryukov:87}
\nocite{Kryukov:87a}
\nocite{Kryukov:88}
\nocite{Kryukov:88a}
\nocite{Kryukov:88b}
\nocite{Kuppers:71}
\nocite{Lambin:84}
\nocite{Lang:79}
\nocite{Laursen:79}
\nocite{Laursen:80}
\nocite{Laursen:81}
\nocite{Lecourtier:85}
\nocite{Lee:85}
\nocite{Leler:85}
\nocite{Lepage:83}
\nocite{Levi:70}
\nocite{Levi:71}
\nocite{Liebermann:75}
\nocite{Liska:84}
\nocite{Liska:87}
\nocite{Liska90}
\nocite{Lloyd:90}
\nocite{Loe:85}
\nocite{London:74}
\nocite{Loos:72}
\nocite{Lottati}
\nocite{Louw:86}
\nocite{Luegger:73}
\nocite{Luegger:91}
\nocite{Lukacs}
\nocite{Lukaszuk:87}
\nocite{Lux:75}
\nocite{MacCallum:86}
\nocite{MacCallum:86a}
\nocite{MacCallum:87}
\nocite{MacCallum:88}
\nocite{MacCallum:89}
\nocite{Mack:73}
\nocite{Mack:73a}
\nocite{Maclaren:89}
\nocite{Maguire:81}
\nocite{Malm:82}
\nocite{Marti:78}
\nocite{Marti:79}
\nocite{Marti:80}
\nocite{Marti:83}
\nocite{Marti:85}
\nocite{Marti:85a}
\nocite{Marti:88}
\nocite{Matveev:87}
\nocite{Maurer:86}
\nocite{Mazepa:85}
\nocite{Mazzarella:85}
\nocite{McCrea:81}
\nocite{McCrea:82}
\nocite{McCrea:83}
\nocite{McCrea:84}
\nocite{REDUCE:84}
\nocite{McCrea:87}
\nocite{McCrea:87a}
\nocite{McCrea:88}
\nocite{McIsaac:85}
\nocite{Melenk:88}
\nocite{Melenk:89}
\nocite{Melenk:89a}
\nocite{Melenk:89b}
\nocite{Mirie:84}
\nocite{Moller:89}
\nocite{Moritsugu:85}
\nocite{Moritsugu:88}
\nocite{Moritsugu:89}
\nocite{Moritsugu:89a}
\nocite{Mueller:81}
\nocite{Murzin:85}
\nocite{Nagata:82}
\nocite{Nagata:85}
\nocite{Nakamura:89}
\nocite{Nakashima:84}
\nocite{Nakashima:84a}
\nocite{Namba:86}
\nocite{Nemeth:82}
\nocite{Nemeth:87}
\nocite{Neun:88}
\nocite{Neun:89}
\nocite{Neun:89a}
\nocite{Neutsch:85}
\nocite{Neutsch:86}
\nocite{Ng:89}
\nocite{Niki:84}
\nocite{Nikityuk:87}
\nocite{Noor:79}
\nocite{Norman:77}
\nocite{Norman:78}
\nocite{Norman:79}
\nocite{Norman:83}
\nocite{Norman90}
\nocite{Norton:80}
\nocite{Nucci:90}
\nocite{Ochiai:90}
\nocite{Ogilvie:82}
\nocite{Ogilvie:89}
\nocite{Ono:1979}
\nocite{Ozieblo}
\nocite{Padget90}
\nocite{Pankau:73}
\nocite{Pankau:73a}
\nocite{Parsons:68}
\nocite{Parsons:71}
\nocite{Pattnaik:83}
\nocite{Pearce:81}
\nocite{Pearce:83}
\nocite{Perjes:84}
\nocite{Perjes:84a}
\nocite{Perjes:84b}
\nocite{Perjes:84c}
\nocite{Perjes:86}
\nocite{Perjes:86a}
\nocite{Perjes:88}
\nocite{Perlt:90}
\nocite{Perrottet:78}
\nocite{Pesic:73}
\nocite{Pictiaw:69}
\nocite{Piessens:84}
\nocite{Piessens:86}
\nocite{Pignataro:85}
\nocite{Podgorzak:84}
\nocite{Price:84}
\nocite{Quarton}
\nocite{Quarton:84}
\nocite{Rao:85}
\nocite{Rayna:87}
\nocite{Reusch:86}
\nocite{Rink:71}
\nocite{Rizzi:85}
\nocite{Rodionov:84}
\nocite{Rodionov:87}
\nocite{Rodionov:87a}
\nocite{Rodionov:88}
\nocite{Rogers:89}
\nocite{Roque:88}
\nocite{Ronveaux:88}
\nocite{Ronveaux:89}
\nocite{Saez:83}
\nocite{Sage:88}
\nocite{Sasaki:79}
\nocite{Savage:90}
\nocite{Sayers:87}
\nocite{Sayers:87a}
\nocite{Schmuck:77}
\nocite{Schruefer:81}
\nocite{Schruefer:82}
\nocite{Schruefer:87}
\nocite{Schruefer:88}
\nocite{Schwarz:80}
\nocite{Schwarz:82}
\nocite{Schwarz:82a}
\nocite{Schwarz:83}
\nocite{Schwarz:83a}
\nocite{Schwarz:84}
\nocite{Schwarz:84a}
\nocite{Schwarz:85}
\nocite{Schwarz:85a}
\nocite{Schwarz:86}
\nocite{Schwarz:87}
\nocite{Schwarz:88}
\nocite{Shablygin:87}
\nocite{Shmueli:83}
\nocite{Shmueli:83a}
\nocite{Shtokhamer:75}
\nocite{Shtokhamer:77}
\nocite{Smit:79}
\nocite{Smit:81}
\nocite{Smit:82}
\nocite{Smit:87}
\nocite{Soderstrand:72}
\nocite{Soderstrand:72a}
\nocite{Soderstrand:74}
\nocite{Soma:77}
\nocite{Soma:85}
\nocite{Spiridonova:87}
\nocite{Squire}
\nocite{Steinberg:82}
\nocite{Steuerwald}
\nocite{Stoutemyer:74}
\nocite{Stoutemyer:75}
\nocite{Stoutemyer:77}
\nocite{Stroscio:74}
\nocite{Stuart:88}
\nocite{Stuart:90}
\nocite{Suppes:89}
\nocite{Surguladze:89}
\nocite{Tallents:84}
\nocite{Tao90}
\nocite{Tasso:76}
\nocite{Thas:89}
\nocite{Thas:89a}
\nocite{Todd:88}
\nocite{Toth:86}
\nocite{Tournier:79}
\nocite{Tournier:87}
\nocite{Trotter:89}
\nocite{Tsai:65}
\nocite{Tsai:74}
\nocite{Ucoluk:82}
\nocite{Umeno:89}
\nocite{vandenHeuvel:86}
\nocite{vandenHeuvel:86a}
\nocite{vandenHeuvel:87}
\nocite{vandenHeuvel:87a}
\nocite{vanHeerwaarden}
\nocite{vanHulzen:80}
\nocite{vanHulzen:81}
\nocite{vanHulzen:82}
\nocite{vanHulzen:82a}
\nocite{vanHulzen:83}
\nocite{vanHulzen:83a}
\nocite{vanHulzen:87}
\nocite{vanHulzen:88}
\nocite{vanHulzen:89}
\nocite{vanHulzen:89a}
\nocite{VanProeyan:76}
\nocite{VanProeyan:79}
\nocite{Vinitsky:87}
\nocite{Voros:77}
\nocite{Wanas}
\nocite{Wanas:85}
\nocite{Wang:84}
\nocite{Wassam:87}
\nocite{Wassam:87a}
\nocite{Watanabe:85}
\nocite{Watanabe:76}
\nocite{Watanabe:79}
\nocite{Weber:79}
\nocite{Wehner:86}
\nocite{Winkelmann:89}
\nocite{Winkler:88}
\nocite{Witham:77}
\nocite{Wood:89}
\nocite{Wright:84}
\nocite{Wulkow:90}
\nocite{Yamamoto:87}
\nocite{Yannouleas:88}
\nocite{Yannouleas:89}
\nocite{Zacrep:75}
\nocite{Zahalak:87}
\nocite{Zeng:84}
\nocite{Zhidkova:78}

\bibliography{bibl}
\bibliographystyle{plain}

\end{document}




